Author's response to reviews

Title: Role of hemoclip in the management of acute bleeding from gastric stromal tumor: A Case Report and review of the literature

Authors:

Mouen A Khashab (mkhashab@iupui.edu)
Harvery M Cramer (hcramer@iupui.edu)
Suthat Liangpunsakul (sliangpu@iupui.edu)

Version: 3 Date: 10 July 2007

Author's response to reviews:

- I added the word A before the 2 words case report in the title
- added a conclusion
- added contributing interests, authors' contributions and acknowledgements
- added the table to the main manuscript